Author's response to reviews

Title: Systematic review and meta-analysis of Transurethral Needle Ablation in symptomatic Benign Prostatic Hyperplasia.

Authors:

Carmen Bouza (cbouza@isciii.es)
Teresa Lopez (teresalc@isciii.es)
Angeles Magro (angelines@isciii.es)
Lourdes Navalpotro (lourdes@isciii.es)
Jose Maria Amate (jamate@isciii.es)

Version: 5 Date: 19 June 2006

Author's response to reviews:

19th June 2006

Dear Annabel Phillips,
Following your indications we have condensed the Tables and Figures and have reorganized the manuscript according to the changes made.
Figure 1 has been deleted
Tables: Table 1 has been deleted. The text is now included in the methods section of the manuscript.
Table 3 has been deleted but relevant data is included in Table 1
Tables 7 and 9 have been deleted
We have included the explanation of RCT in Table 6 (now Table 4).
In addition, we have included a competing interest section. We hope you will find these changes to your liking, and we will look forward to hearing from you.
Yours sincerely,
Carmen Bouza, MD, PhD
Agency for Health Technology Assessment. Instituto de Salud Carlos III.
Sinesio Delgado 4, 28029 Madrid, Spain.
Telephone number:+34 91 822 2018
Fax: +34 91 387 7841.
e-mail: cbouza@isciii.es